Author's response to reviews

Title: Macular thickness measurements in healthy Norwegian volunteers: an optical coherence tomography study

Authors:

Alexandra Wexler (wexler@ntnu.no)
Trond Sand (Trond.sand@ntnu.no)
Tor B Elsås (Elsas@ntnu.no)

Version: 7 Date: 5 May 2010

Author's response to reviews:

Dear Editorial Team. Three small adjustments were maid in this final revision:

1. "minimal foveal" was changed to "minimal foveolar" in Abstracts section under "results". Same change was done in Abstracts on the website.

2. One hard spacing was added in same section to separate subsection.

3. "Optical Coherence Tomography" was changed to "optical coherence tomography", removing capitals.

Thank you again for your help!

The best

Alexandra Wexler